Author's response to reviews

Title: Influenza in long-term Dutch travelers in the tropics: symptoms and infections

Authors:

Jane Whelan (whelanjane@hotmail.com)
Guus F Rimmelzwaan (g.rimmelzwaan@erasmusmc.nl)
Anneke van den Hoek (vandenhoek_anneke@yahoo.co.uk)
Sanne M Belderok (belderoksanne@gmail.com)
Gerard J.B Sonder (gsonder@ggd.amsterdam.nl)

Version: 3
Date: 5 February 2016

Author's response to reviews: see over
Amsterdam, 5 February 2016
Subject: MS: 1726616506173494 - Influenza in long-term Dutch travelers in the tropics: symptoms and infections

Dear editor,

My co-authors and I would like to thank the reviewers again for their time and we are happy that our article has now been accepted for publication.

In reply to your editorial comments:

Editorial comments:

1. Conclusions
Please add a section Conclusions after section Discussion, which states clearly the main conclusions and includes an explanation of their relevance or importance to the field. You may wish to use the last paragraph of Discussion for that purpose.

We have now, as you suggest, added a section conclusions and used the last paragraph of the discussion for this purpose.

2. Author’s Contributions
We would appreciate if the initials are used instead of the full names, and if a statement is added that confirms that all authors read and approved the final version of the manuscript.

We have now used initials instead of full names and have confirmed that all authors read and approved the final version of the manuscript.

3. Figure
Please state in the cover letter whether the map depicted in your figure is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

The figure with the map has been drawn by Floor Elfrink, who we acknowledge for this in the acknowledgement section. It is our own and it is not under copyright.

4. Manuscript document
At this stage we would appreciate if the manuscript document and all related files do not have any tracked changes, comments, highlights or text in different colours.

We have now removed all tracked changes, comments and highlights before resubmission.

Best regards,
Jane Whelan